Author's response to reviews

Title: Calibrating ADL-IADL scales to improve measurement accuracy and to extend the disability construct into the preclinical range: a systematic review

Authors:

Robert A Fieo (r.fieo@sms.ed.ac.uk)
Elizabeth J Austin (elizabeth.austin@ed.ac.uk)
John M Starr (jstarr@staffmail.ed.ac.uk)
Ian J Deary (iand@staffmail.ed.ac.uk)

Version: 6 Date: 30 July 2011

Author's response to reviews:

MAJOR revisions:
1)'Robert Fieo' changed to Robert A Fieo
2)text abstract made identical to the abstract within the submission system, i.e., changed bold font to normal
3)references were re-numbered, so that references that are cited in table 1 are last numerically
4)Table 2 was collapsed to make one table, rather than the existing arrangement which was table 2 and table 2 continued.

MINOR revision
1)Label 'additional files' changed to 'additional files 1'
2)Figure 1 cropped to reduce white space